



https://helda.helsinki.fi

Longitudinal analysis of risk of non-alcoholic fatty liver disease in adulthood

Cuthbertson, Daniel J.

2019-06

Cuthbertson, D J, Brown, E, Koskinen, J, Magnussen, C G, Hutri-Kähönen, N, Sabin, M, Tossavainen, P, Jokinen, E, Laitinen, T, Viikari, J, Raitakari, O T & Juonala, M 2019, 'Longitudinal analysis of risk of non-alcoholic fatty liver disease in adulthood ', Liver International, vol. 39, no. 6, pp. 1147-1154. https://doi.org/10.1111/liv.13993

http://hdl.handle.net/10138/306232 https://doi.org/10.1111/liv.13993

acceptedVersion

Downloaded from Helda, University of Helsinki institutional repository.

This is an electronic reprint of the original article.

This reprint may differ from the original in pagination and typographic detail.

Please cite the original version.

Article type : Original Articles

Editor : Luca Valenti

Longitudinal analysis of risk of Non-Alcoholic Fatty Liver Disease (NAFLD) in

adulthood.

Daniel J Cuthbertson,¹ Emily Brown,¹ Juha Koskinen,^{2,3} Costan G. Magnussen,^{2,4} Nina Hutri-Kähönen⁵ Matthew Sabin^{6,7} Päivi Tossavainen⁸ Eero Jokinen⁹ Tomi Laitinen¹⁰ Jorma Viikari^{11,12} Olli T Raitakari² Markus Juonala^{6,11,12}

¹Obesity and Endocrinology Research Group, Institute of Ageing and Chronic Disease, University of Liverpool, UK,

²Research Centre of Applied and Preventive Cardiovascular Medicine, University of Turku, Finland

³Heart Center, Kymenlaakson keskussairaala, Kotka, Finland ⁴Menzies Institute for Medical Research, University of Tasmania, Hobart, Australia ⁵Department of Pediatrics, University of Tampere and Tampere University Hospital, Tampere, Finland ⁶Murdoch Children's Research Institute, Melbourne, Victoria, Australia ⁷The Royal Children's Hospital and University of Melbourne, Melbourne, Victoria, Australia

⁸Department of Pediatrics, PEDEGO Research Unit and Medical Research Center, Oulu University Hospital and University of Oulu, Oulu, Finland. ⁹Department of Pediatric Condicional Medical Condicional Adolescente, University of

⁹Department of Pediatric Cardiology, Hospital for Children and Adolescents, University of Helsinki, Helsinki, Finland.

¹⁰Department of Clinical Physiology and Nuclear Medicine, Kuopio University Hospital and University of Eastern Finland, Kuopio, Finland.

¹¹Department of Medicine, University of Turku, Turku, Finland. ¹²Division of Medicine, Turku University Hospital, Turku, Finland.

Corresponding author & address for reprints: Dr Daniel Cuthbertson

Clinical Sciences Centre, Aintree University Hospital, Liverpool, L9 7AL

E-mail: daniel.cuthbertson@liverpool.ac.uk

Tel 0151 529 5911,

Fax 0151 529 5888

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1111/liv.13993

Abbreviations: NAFLD, non-alcoholic fatty liver disease; BMI, body mass index; HDL, high-density lipoprotein; ALT, alanine aminotransferase; AST, aspartate aminotransferase

Conflict of Interest The authors who have taken part in this study declared that they do not have anything to disclose regarding funding or conflict of interest with respect to this manuscript

Funding The Young Finns Study has been supported by Academy of Finland grants 126925, 121584, 124282, 129378 (Salve), 117797 (Gendi), and 41071 (Skidi); the Social Insurance Institution of Finland; Kuopio, Tampere, and Turku University Hospital Medical Funds (grants 9M048 and 9N035); the Juho Vainio Foundation; the Paavo Nurmi Foundation; the Finnish Foundation for Cardiovascular Research; the Finnish Cultural Foundation; the Tampere Tuberculosis Foundation; and the Emil Aaltonen Foundation. This work was also funded by the National Health and Medical Research Council (grant APP1098369). C.G.M is supported by a National Heart Foundation of Australia Future Leader Fellowship (100849). J.K. was supported by Turku University Foundation, Emil Aaltonen Foundation and Urmas Pekkala Foundation

Abstract

Background & Aims We aimed to determine how childhood body mass index (BMI) and metabolic health, along with change in BMI between childhood and adulthood, determine the risk of adult non-alcoholic fatty liver disease (NAFLD).

Methods Data from 2,020 participants aged 3-18 years at baseline, followed up 31 years later,

was examined to assess the utility of four childhood metabolic phenotypes (metabolic groups I: normal BMI, no metabolic disturbances; II: normal BMI, one or more metabolic disturbances; III: overweight/obese, no metabolic disturbances; IV: overweight/obese, one or more metabolic disturbances) and four life-course adiposity phenotypes (adiposity group 1: normal child and adult BMI; 2, high child, normal adult BMI; 3, normal child BMI, high adult BMI; 4, high child and adult BMI) in predicting adult NAFLD.

Results The risk for adult NAFLD was similar across all four groups after adjustment for age, sex, lifestyle factors and adult BMI. Risk of adult NAFLD was not increased among individuals overweight/obese in childhood but non-obese in adulthood. In contrast, overweight or obese adults, irrespective of their youth BMI status, had ~8-10 fold increased risk (P<0.001).

Conclusion Childhood overweight/obesity, not metabolic health, is associated with increased risk for adult non-alcoholic fatty liver disease. However, the increased risk associated with childhood overweight/obesity can be largely removed by obtaining a normal body mass index by adulthood.

Abstract Keywords Non-alcoholic fatty liver disease, metabolic health, obesity, risk

Lay Summary

Non-alcoholic fatty liver disease or '*fatty liver*' is usually seen in people who are overweight/obese and is associated with other metabolic complications (high blood pressure, high cholesterol) and long-term health problems. Using data from the Cardiovascular Risk in Young Finns study we assessed how body weight (lean/overweight/obese) or having metabolic complications during childhood increases the risk of developing fatty liver in

adulthood. Obesity, rather than metabolic complications alone, is the major risk factor for developing fatty liver. However, individuals who were overweight/obese during childhood but became lean as adults do not have a higher risk of fatty liver in adulthood.

Introduction

Non-alcoholic fatty liver disease (NAFLD) is the most common form of chronic liver disease in Western society with a recent meta-analysis of 8.5 million people from 22 countries suggesting a global prevalence of NAFLD of 25.2.¹ The prevalence of NAFLD is even higher in selected populations, such as individuals with components of the metabolic syndrome and type 2 diabetes.^{2,3} Conversely, in patients diagnosed with NAFLD, obesity, type 2 diabetes and metabolic syndrome co-exist in 51%, 22.5% and 42.5% of cases.¹ Thus, NAFLD has been described as the hepatic manifestation of the metabolic syndrome.⁴

Obesity is the major risk factor for NAFLD. However, while many overweight or obese individuals have evidence of metabolic complications (hypertension, dyslipidaemia, insulin resistance), a proportion may not have such complications *i.e.* are metabolically healthy.⁵ In the UK BioBank cohort, in which liver fat was characterised by magnetic resonance spectroscopy in 4,989 individuals, the proportion of overweight/obese individuals with normal liver fat was two-fold higher than that of overweight/obese individuals with higher liver fat (42.9 *vs.* 18.1%).⁶ Liver fat tends to be associated with poor metabolic health such that metabolically unhealthy individuals, whether lean or obese, tend to have greater liver fat.^{5,7}

Although the concept of metabolically healthy obesity is contentious, a number of studies have examined the association of metabolically healthy obesity with incident heart failure, cardiovascular disease and type 2 diabetes with conflicting results.⁸⁻¹³ Few studies have examined the association between metabolic health status and obesity with incident NAFLD.¹⁴⁻¹⁶ These analyses have been performed among adult populations. Thus, whether it is childhood weight status, or the associated metabolic phenotype, that increases the risk of NAFLD remains unclear.

The main aim of this study was to assess the relative ability of childhood (age 3-18 years) weight status (normal, overweight or obese) and presence or absence of metabolic complications (i.e. being metabolically healthy *vs.* unhealthy), in a large, community-based cohort of young adults, to determine the relative risk of developing NAFLD in adulthood. In addition, we aimed to evaluate the combined effects of child and adult weight status to assess whether the influence of child overweight or obesity is reversible.

Patients and Methods

Participants and Study Design

The Cardiovascular Risk in Young Finns Study is an ongoing multicentre study examining precursors of atherosclerosis in children and adolescents. It was first launched in 1980 when 3,596 participants, aged between 3-18 (3, 6, 9, 12, 15 and 18) years, were randomly selected from the national register, across Finland, to produce a representative sample of Finnish children. Thereafter, follow-up studies have been conducted regularly in 3-year interval until 1992. The most recent 31-year follow-up (2011) among 2,042 participants included clinical examination, blood samples, questionnaires and liver ultrasound examinations.

The sample for the present analysis included those who had participated in the 1980 survey (baseline assessment) and who had liver ultrasound examinations performed in 2011. We have previously shown that participants who attended follow-up studies shared characteristics similar to those who attended at baseline.^{17,18} We included in our analysis the following baseline characteristics: age, gender, BMI, blood pressure, glucose, high-density lipoprotein (HDL)-cholesterol and triglycerides concentrations. For the follow-up in 2011, we included age, gender, BMI, blood pressure, glucose, alanine aminotransferase (ALT), aspartate aminotransferase (AST), HDL-cholesterol and triglycerides concentrations. In addition, we included data on alcohol use, diet, physical activity and liver ultrasound results. During the follow up period there have been 110 deaths recorded: 36 deaths from endogenous causes and 74 deaths from external causes (Supplementary Figure 1). The study was approved by local ethics committees of Turku, Tampere, Oulu, Kuopio and Helsinki, and written informed consent was obtained from all of the study participants or their parents if they were under 18 years of age.

Clinical Measurements

In childhood and adulthood, height and weight were measured and body mass index (BMI) was calculated as weight in kilograms divided by height in metres squared. Childhood BMI was adjusted for age and sex by calculating a BMI-SD score. Blood pressure was measured from the brachial artery with a standard sphygmomanometer with cuff size chosen according to arm circumference.

Questionnaires were used to obtain data on smoking and physical activity. The physical activity index was calculated as previously described.¹⁹ Adulthood alcohol consumption data were acquired by standardised questionnaires. Alcohol intake was defined as consumed alcohol units per week (16 g of alcohol). Information on dietary habits was obtained with a

detailed food frequency questionnaire providing an estimate of food consumption in grams/day.

Data collected from national hospital discharge registries confirmed that none of the participants had viral or auto-immune liver disease.

Biochemical Measurements

Venous blood samples were drawn after an overnight fast for determination of serum lipid levels, insulin and CRP. Serum insulin was measured with immunoassay. All lipid and lipoprotein determinations were performed on serum using standard methods, as described previously.^{18,20} Low density lipoprotein cholesterol concentration was calculated by the Friedewald formula in participants with triglycerides <4.0 mmol/l.

Genetic Analysis

In the present study, we used the SNPs rs738409, near the patatin-like phospholipase domaincontaining protein 3 (*PNPLA3*) gene and rs58542926, near the transmembrane 6 superfamily member 2 (*TM6SF2*) gene associated with fatty liver in previous genome-wide association analyses as the genetic marker for susceptibility for fatty liver, as previously described.²¹ Genotyping was performed with the custom-built Illumina BeadChip 670K.

Ultrasound Measurements

Presence of hepatic steatosis NAFLD was determined by ultrasound imaging of the liver using a validated protocol and Sequoia 512 ultrasound mainframes (Acuson, Mountain View, CA, USA) with 4.0 MHz adult abdominal transducers.²² The evaluation of fatty liver was done by one ultrasonographer, and the clinical diagnosis (fatty liver or normal liver) was based on visual assessment of the hepatic-renal ratio, liver parenchymal brightness and the visibility of vessel walls and the diaphragm.²¹

Classification of Metabolic Disturbances, Metabolic Syndrome and Overweight at Youth

In the absence of diagnostic criteria for paediatric metabolic abnormalities, we generated ageand sex-specific percentiles of systolic and diastolic blood pressures, HDL-cholesterol, triglyceride, and glucose levels from prevalence data from the National Cholesterol Education Program, Report of the Expert Panel on Blood Cholesterol Levels in Children and Adolescents as previously described.²² Participants were categorised as having a metabolic disturbance in youth if he/she had any of the following: systolic blood pressure $\geq 90^{\text{th}}$ percentile and/or diastolic blood pressure $\geq 90^{\text{th}}$ percentile, insulin $\geq 90^{\text{th}}$ percentile, triglycerides $\geq 90^{\text{th}}$ percentile, or HDL-cholesterol $\leq 10^{\text{th}}$ percentile.²²

Participants were categorised into four metabolic groups on the basis of i) *youth BMI category* using age- and sex-specific international BMI centiles to extrapolate cut-points at ages 3 to 18 that correspond to adult BMI of 25 kg/m² (overweight) and 30 kg/m² (obese)²³ and ii) *metabolic status*. *Metabolic Group I*, the reference group, included participants with normal BMI and no metabolic disturbances *i.e.* metabolically healthy; *Metabolic Group II* included participants with normal BMI, but who had one or more metabolic disturbances, *i.e.* metabolically unhealthy; *Metabolic Group* III included participants who were overweight/obese with no metabolic disturbances, *i.e.* metabolically healthy; and *Metabolic Group* IV included participants who were overweight/obese, with one or more metabolic disturbances *i.e.* metabolically unhealthy.

Classification of Adiposity Groups

Participants were grouped into four separate adiposity groups as previously described²⁴ according to their BMI in childhood and as adults. For the definitions of childhood overweight and obesity, we used age-specific and sex-specific international BMI percentiles to extrapolate cut-off points for those aged 3 to 18 years that correspond to adult BMIs of 25 (overweight) and 30 kg/m² (obese).²³

The groupings were as follows: *Adiposity Group 1*, participants with a normal BMI in childhood and normal BMI as adults; *Adiposity Group 2*, participants who were overweight or obese in childhood but had a normal BMI as adults; *Adiposity Group 3*, participants with a normal BMI in childhood who were overweight as adults and *Adiposity Group 4*, participants who were overweight or obese in childhood and overweight as adults.

Statistical Methods

Statistical analyses were performed with SAS 9.4. Statistical significance was inferred as a P value of less than 0.05. The normality assumptions of the residuals were assessed by examining histograms of the residuals and normal probability plots. The residuals were normally distributed. Values for plasma triglycerides and alcohol consumption were log transformed to correct for skewness. Baseline and clinical characteristics across each of the metabolic/BMI groups are presented as mean±SD for continuous variables or percent for categorical variables. ANOVA was used to compare characteristics among the study groups. Age and sex adjusted logistic regression model was used to calculate odds ratios and 95% confidence intervals (model 1 in the text). Further adjustments were done by including adult lifestyle habits (model 2 in the text) and adult BMI (model 3 in the text) as covariates in the models.

Thereafter additional adjustments for adulthood alcohol consumption (portions per week), smoking (yes/no) and physical activity, and finally also for adulthood BMI were performed.

Results

Participant Characteristics

Participant characteristics are provided in Table 1, sub-divided into four metabolic groups according to their baseline BMI (normal *vs.* overweight/obese) and the number of associated metabolic risk factors (none *vs.* \geq 1). The proportion of participants with NAFLD in adulthood for each group was Metabolic Group I: 16.9%, Group II: 20.4%, Group III: 26.6%, and Group IV: 40.5%. We observed no difference in BMI change from childhood to adulthood between Group I and IV. Normal weight was gained by 9 participants (11 %) in the metabolic group III and 12 participants (15 %) in the metabolic group IV during the follow-up from childhood to adulthood. Details of BMI change measured from two time points has been presented in the online supplement (Supplementary Table 1).

Association of Youth BMI Status and Metabolic Health on Adult NAFLD

The risk of adult NAFLD compared according to their youth phenotype is shown in Table 2. The odds ratios of NAFLD were significantly greater in overweight/obese participants (Metabolic group III and IV) compared with normal BMI participants (Metabolic group I), with the highest odds ratio among the metabolically unhealthy, overweight/obese group (Metabolic group IV). These differences were evident after adjustment for adult alcohol consumption (portions per week), smoking (yes/no) and physical activity (Table 2, Model 2). However, when adjusted for adult BMI, the odds ratios between all four groups were not significantly different from the reference group (Table 2, Model 3). Age-stratified analyses (Supplementary Table 2) were largely consistent with those shown in Table 2. We also

compared those with normal BMI but metabolic disturbances (Metabolic group II) with those who were overweight or obese but metabolically healthy (Metabolic group III). No significant differences between the groups were observed (age and sex adjusted OR 1.53 95 % CI 0.90-2.62, P=0.11). When comparing participants with normal BMI but metabolic disturbances (Metabolic group II) with those who were overweight or obese and had metabolic disturbances, we observed a significantly higher odds ratio for adult NAFLD between the groups (age and sex adjusted OR 2.54 95 % CI 1.51-4.29, P=0.0005). However, this association was not observed after additional adjustment for adult NAFLD when those overweight or obese but metabolically healthy youth (metabolic group III) were compared with those who were overweight or obese and had metabolic group III) healthy youth (metabolic group III) were compared with those who were overweight or obese and had metabolic disturbances (metabolically healthy youth (metabolic group III) were compared with those who were overweight or obese and had metabolic disturbances (metabolic group IV) (data not shown).

Association of Child Metabolic Syndrome on Risk of Adult NAFLD

We examined the effect of child metabolic syndrome status at baseline on the subsequent risk of developing NAFLD in adulthood. The odds ratio of subsequent NAFLD was higher in those with childhood metabolic syndrome versus those with no childhood metabolic syndrome, even after adjustment for age and sex (Table 3, Model 1), adulthood alcohol consumption, smoking and physical activity (Table 3, Model 2). However, after further adjustment for adult BMI, there was no difference in risk of NAFLD between those with or without child metabolic syndrome (Table 3, Model 3).

Association of BMI Transition from Childhood to Adulthood on Risk of Adult NAFLD

When using a BMI cut-off of 25 kg/m², participants who were overweight/obese as adults (adiposity groups 3 and 4) had approximately 8-fold increased odds ratio of developing NAFLD, compared to the normal BMI group (group 1), regardless of childhood adiposity status (Table 4A). Contrary, those who were not overweight/obese in adulthood (group 2) had similar odds ratio for NAFLD compared to the normal BMI group (group 1). If a BMI cut-off of 30 kg/m² was applied, similar results were found: participants who were obese (adiposity groups 3 and 4) had an 8-10 fold increased odds of developing NAFLD, compared with the normal BMI group (group 1), regardless of childhood adiposity status (Table 4B). Contrary, those who were not obese as adults (group 2) had similar risk for NAFLD compared with the normal BMI group (group 1).

Discussion

The results of this study demonstrate that obesity status in childhood, independent of metabolic health, is predictive of NAFLD, suggesting measurement of body mass index, rather than that of associated metabolic risk factors, provides sufficient information for risk assessment of NAFLD. With the strong tracking of BMI from childhood to adulthood, obesity seems to be the major factor in the development of adulthood NAFLD although reversal of childhood overweight/obesity status in adulthood was not associated with an increased risk of NAFLD suggesting childhood BMI as a modifiable risk factor for NAFLD.

We have previously examined individual childhood predictors of adult fatty liver using the Young Finns Study cohort and demonstrated being small for gestational age, high serum insulin, high BMI and genetic variants in PNPLA3 and TM6SF2 as being important risk factors.²¹ We have also demonstrated that metabolic aberrations appear to precede the development of fatty liver in young adults, up to 10 years before diagnosis. The most striking

metabolic abnormalities related to high concentrations of triglyceride-related lipid measures and branched chain amino acids.²⁵ The latest findings further extend our understanding of the contributory factors for development of NAFLD in adulthood.

The findings of the present study provide important novel information concerning the combined effects of childhood obesity and metabolic risk factors on adult NAFLD. We evaluated the effects of metabolically healthy vs. unhealthy obesity and explored whether the effects of childhood obesity and metabolic factors are mediated by adult obesity. These data strongly support the premise that it is obesity rather than metabolic health as the main driver for NAFLD. Initial assessments of the relative risk of NAFLD according to the four different metabolic groups, suggested metabolic phenotype was relevant, with the greatest odds ratio in the metabolically unhealthy obese, even after adjustment for age, sex, alcohol, smoking and physical activity. However, after adjustment for adulthood BMI any differences in relative risk were no longer evident. Consistent with these data, the presence or absence of childhood metabolic syndrome did not influence the relative risk of NAFLD when adulthood BMI status was considered. In contrast, those who were overweight/obese (whether a BMI cut-off of 25 or 30 kg/m² was adopted) in adulthood had a significantly greater increased risk of NAFLD, compared to the normal BMI population, regardless of childhood adiposity status. The relative risk was greater in overweight/obese individuals who had been overweight/obese as children than in those individuals who had been of normal BMI in childhood.

Significantly, the risk of NAFLD among overweight or obese children, who became nonobese by adulthood, was similar to those who were never obese, and this finding is consistent with our previous findings for the risks of other cardio-metabolic conditions.²⁴ We have previously shown similar risks of type 2 diabetes, hypertension, dyslipidaemia and carotidartery atherosclerosis in overweight or obese children who became non-obese by adulthood. Many studies examining obesity and its associated risks, examine the dose-response

relationship between BMI and hazard ratios/incidence rates of the complications overlooking the impact of the duration of obesity. Duration of obesity is an independent risk factor for obesity-related complications²⁶ and data from the Framingham Heart Study reveal duration of obesity to also be a relevant risk factor for NAFLD with an incremental risk for every additional year of obesity. In this study, although we were unable to examine the duration of obesity in years for any given individual, we were able to demonstrate that children who were overweight or obese in childhood and who were of normal BMI when re-examined in adulthood were not more likely to have NAFLD.

There have been several studies performed to date among adult populations that have attempted to address the same question as the one posed in this study. Sung et al. undertook a cross-sectional assessment of 14,384 South Koreans and found that metabolically healthy obese individuals were at risk of fatty liver but attenuated risk of pre-clinical atherosclerosis, although this study did not address the relative contributions of weight versus metabolic factors in development of NAFLD.¹⁴ Chang et al. examined 77,425 individuals, who were free of NAFLD or associated metabolic abnormalities at baseline, and re-examined them after 4.5 years of follow up, found an adjusted hazards ratio of 2.15 (2.06-2.26) and 3.55 (3.37-3.74) for developing NAFLD in overweight and obese respectively, compared with normal BMI, individuals.¹⁵ The association between BMI and hazard ratio of NAFLD was progressive across the whole range of BMI values suggesting that it is obesity per se, independent of co-existing insulin resistance or metabolic abnormalities, that increases the risk of NAFLD. In contrast, Lee et al., examined 3,045 individuals without NAFLD and diabetes at baseline and re-evaluated them after 4 years. Their findings of a lower risk of NAFLD in metabolically healthy obese versus metabolically unhealthy non-obese led to the conclusion that metabolic health is more important than obesity in the development of NAFLD.¹⁶ However, contrary to the present analyses, their study cohort consisted of adults (mean age 44 years).

The strengths of this study are the large, randomly selected cohort, who have undergone careful metabolic characterisation, have been followed up after a significant interval of 31 years and with their initial assessment occurring in childhood/early adulthood. The availability of accurate information on lifestyle factors, particularly alcohol intake and physical activity patterns that would otherwise confound analysis is also advantageous.

We must acknowledge certain limitations to this study. First, we did not have data on liver enzymes measured in childhood. Secondly, because our study cohort was a homogeneous ethnic group, the generalisability of our results is limited to Caucasians. Thirdly, the loss of original participants during the long-term follow-up is also recognised. Non-participants were younger and more often male in childhood than participants. Therefore, the rates of adult fatty liver in our cohort might be an underestimation of the real rates. However, a detailed analysis of participants who dropped out was performed after the 21-year follow-up study in 2001 (n=2283) and the findings suggest the present study cohort is representative of the original study population.¹⁸ The application of ultrasonography to detect incident fatty liver will underestimate the true prevalence of NAFLD. Although ultrasound has been validated against the gold standard method of liver fat quantification of magnetic resonance spectroscopy, it is generally accepted it will detect only moderate or severe steatosis and mild steatosis will not be detectable.²⁷ Large scale epidemiological studies have applied these gold-standard measures but with obvious implications of cost, feasibility and resource availability.^{28,29} Furthermore, no information is available from liver ultrasound on the staging (steatosis, steatohepatitis or fibrosis etc.) within the NAFLD spectrum. Future research directions for this cohort will be to apply non-invasive biomarkers of liver fibrosis (e.g. using the Fib 4 or NAFLD fibrosis score) or even further imaging (e.g. using elastography) to

assess for fibrosis.³⁰ Concerning the changes in BMI status between child and adult measurements, due to lack of comprehensive serial data, we were unable to examine in detail the timing of BMI normalisation. We observed that the influence of child overweight or obesity was reversible with respect to NAFLD. However, it is possible to have different associations in certain genetic groups based on e.g. PNPLA3 genotype. Within our cohort the number of subjects in several PNPLA3/adiposity subgroups did not allow us to conduct properly powered analyses to investigate this.

In conclusion, we demonstrate that obesity, and not the presence of obesity-related metabolic abnormalities, is the overwhelming risk factor for the development of NAFLD in adulthood. The risks are greatest if obesity has been evident since childhood but can be minimised, with risks of NAFLD returning to baseline, if childhood obesity is addressed and the individual returns back to normal BMI in adulthood. Weight loss, through lifestyle intervention, must remain the therapeutic priority for prevention of development and progression of NAFLD.

References

- 1. Younossi ZM, Koenig AB, Abdelatif D, Fazel Y, Henry L, Wymer M. Global epidemiology of nonalcoholic fatty liver disease-Meta-analytic assessment of prevalence, incidence, and outcomes. *Hepatology* 2016;64:73-84.
- 2. Kotronen A, Juurinen L, Hakkarainen A, et al. Liver fat is increased in type 2 diabetic patients and underestimated by serum alanine aminotransferase compared with equally obese nondiabetic subjects. *Diabetes Care* 2008;31:165-169.
- Williamson RM, Price JF, Glancy S, et al. Prevalence of and risk factors for hepatic steatosis and nonalcoholic Fatty liver disease in people with type 2 diabetes: the Edinburgh Type 2 Diabetes Study. *Diabetes Care* 2011;34:1139-1144.

- 6. 7.
- 4. Marchesini G, Brizi M, Bianchi G, et al. Nonalcoholic fatty liver disease: a feature of the metabolic syndrome. *Diabetes* 2001;50:1844-1850.
 - 5. Stefan N, Kantartzis K, Machann J, et al. Identification and characterization of metabolically benign obesity in humans. *Arch Intern Med* 2008;168:1609-1616.
 - . Wilman HR, Kelly M, Garratt S, et al. Characterisation of liver fat in the UK Biobank cohort. *PLoS One* 2017;12:e0172921.
 - Dobson R, Burgess MI, Sprung VS, et al. Metabolically healthy and unhealthy obesity: differential effects on myocardial function according to metabolic syndrome, rather than obesity. *Int J Obes (Lond)* 2016;40:153-161.
 - 8. Morkedal B, Vatten LJ, Romundstad PR, Laugsand LE, Janszky I. Risk of myocardial infarction and heart failure among metabolically healthy but obese individuals. The HUNT Study, Norway. *J Am Col Cardiol* 2013;63:1071-1078.
 - 9. Twig G, Afek A, Derazne E, et al. Diabetes risk among overweight and obese metabolically healthy young adults. *Diabetes Care* 2014;37:2989-2995.
 - Franssens BT, van der Graaf Y, Kappelle JL, et al. Body Weight, Metabolic Dysfunction, and Risk of Type 2 Diabetes in Patients at High Risk for Cardiovascular Events or With Manifest Cardiovascular Disease: A Cohort Study. *Diabetes Care* 2015;38:1945-51.
 - 11. Aung K, Lorenzo C, Hinojosa MA, Haffner SM. Risk of developing diabetes and cardiovascular disease in metabolically unhealthy normal-weight and metabolically healthy obese individuals. *J Clin Endocrinol Metab* 2014;99:462-468.
 - 12. Hamer M, Stamatakis E. Metabolically healthy obesity and risk of all-cause and cardiovascular disease mortality. *J Clin Endocrinol Metab* 2012;97:2482-2488.
 - 13. Ortega FB, Lee DC, Katzmarzyk PT, et al. The intriguing metabolically healthy but obese phenotype: cardiovascular prognosis and role of fitness. *Eur Heart J* 2013;34:389-397.
 - 14. Sung KC, Cha SC, Sung JW, So MS, Byrne CD. Metabolically healthy obese subjects are at risk of fatty liver but not of pre-clinical atherosclerosis. *NMCD* 2014;24:256-262.

- Chang Y, Jung HS, Cho J, et al. Metabolically Healthy Obesity and the Development of Nonalcoholic Fatty Liver Disease. *Am J Gastroenterol* 2016;111:1133-1140.
- Lee MK, Rhee EJ, Kim MC, et al. Metabolic Health Is More Important than Obesity in the Development of Nonalcoholic Fatty Liver Disease: A 4-Year Retrospective Study. *Endocrinol Metab (Seoul, Korea)* 2015;30:522-530.
- Koskinen J, Kahonen M, Viikari JS, et al. Conventional cardiovascular risk factors and metabolic syndrome in predicting carotid intima-media thickness progression in young adults: the cardiovascular risk in young Finns study. *Circulation* 2009;120:229-236.
- Juonala M, Viikari JS, Hutri-Kahonen N, et al. The 21-year follow-up of the Cardiovascular Risk in Young Finns Study: risk factor levels, secular trends and eastwest difference. *J Intern Med* 2004;255:457-468.
- Telama R, Viikari J, Valimaki I, et al. Atherosclerosis precursors in Finnish children and adolescents. X. Leisure-time physical activity. *Acta Paediatr Scand Suppl* 1985;318:169-180.
- Porkka KV, Raitakari OT, Leino A, et al. Trends in serum lipid levels during 1980-1992 in children and young adults. The Cardiovascular Risk in Young Finns Study. Am J Epidemiol 1997;146:64-77.
- Suomela E, Oikonen M, Pitkanen N, et al. Childhood predictors of adult fatty liver. The Cardiovascular Risk in Young Finns Study. *J Hepatol* 2016;65:784-790.
- 22. Koskinen J, Magnussen CG, Sabin MA, et al. Youth overweight and metabolic disturbances in predicting carotid intima-media thickness, type 2 diabetes, and metabolic syndrome in adulthood: the Cardiovascular Risk in Young Finns study. *Diabetes Care* 2014;37:1870-1877.
- Cole TJ, Bellizzi MC, Flegal KM, Dietz WH. Establishing a standard definition for child overweight and obesity worldwide: international survey. *BMJ* 2000;320:1240-1243.
- 24. Juonala M, Magnussen CG, Berenson GS, et al. Childhood adiposity, adult adiposity, and cardiovascular risk factors. *N Engl J Med* 2011;365:1876-1885.

- Kaikkonen JE, Wurtz P, Suomela E, et al. Metabolic profiling of fatty liver in young and middle-aged adults: Cross-sectional and prospective analyses of the Young Finns Study. *Hepatology* 2017;65:491-500.
- 26. Abdullah A, Wolfe R, Stoelwinder JU, et al. The number of years lived with obesity and the risk of all-cause and cause-specific mortality. *Int J Epidemiol* 2011;40:985-996.
- 27. Williamson RM, Perry E, Glancy S, et al. The use of ultrasound to diagnose hepatic steatosis in type 2 diabetes: intra- and interobserver variability and comparison with magnetic resonance spectroscopy. *Clin Radiol* 2011;66:434-439.
- Wong VW, Wong GL, Yeung DK, et al. Incidence of non-alcoholic fatty liver disease in Hong Kong: a population study with paired proton-magnetic resonance spectroscopy. *J Hepatol* 2015;62:182-189.
- 29. Doycheva I, Cui J, Nguyen P, et al. Non-invasive screening of diabetics in primary care for NAFLD and advanced fibrosis by MRI and MRE. *Aliment Pharmacol Ther* 2016;43:83-95.
- Angulo P, Hui JM, Marchesini G, et al. The NAFLD fibrosis score: a noninvasive system that identifies liver fibrosis in patients with NAFLD. *Hepatology* 2007;45:846-854.

Table 1 Clinical characteristics of participants at baseline (1980) and follow up (2011).

Metabolic Group	Metabolic Group I	Metabolic Group	Metabolic Group	Metabolic Group IV	P-value*
	(Reference)	II	III	(Overweight/obese, ≥1	
	(Normal BMI, no	(Normal BMI, ≥1	(Overweight/obese,	metabolic factor)	
	metabolic factors)	metabolic factor)	no metabolic factors)		
Baseline data 1980	N=1455	N=402	N=79	N=84	
Age (years)	10.7+5.1	11 5+4 7	103+44	12 1+4 1	0.005
Sex (male)	46%	44%	44%	51%	0.68
$BMI (kg/m^2)$	17.4+2.6	18.0+2.7	21.9+3.2	23.7+3.9	<0.0001
Systolic blood pressure	111±11	116 ± 12	115 ± 10	122 ± 10	< 0.0001
(mmHg)		-			
Diastolic blood pressure	68±9	69±10	71±8	73±9	< 0.0001
(mmHg)					
Triglycerides (mmol/l)	0.58±0.20	0.88 ± 0.40	0.61±0.22	1.12±0.51	< 0.0001
UDL abolastarol	1 62+0 20	1 28+0 21	1 61+0 27	1 25+0 22	<0.0001
(mmol/l)	1.02±0.27	1.50±0.51	1.01±0.27	1.35±0.35	<0.0001
Insulin (IU/L) (LOGe	6.5+2.9	10.7+1.9	8.9+1.7	15.6+1.8	< 0.0001
corrected values		1007 = 107	000 = 111	1010_110	
 Genetic data					
Variant in PNPLA3	59/36/5	58/36/6	63/36/1	53/41/6	0.57
(CC/CG/GG)%					
Variant in TM6SF2	89/11/0	89/11/0	83/17/0	81/18/1	0.03
(CC/CT/TT)%					
Follow-up data 2011		• •			0.0001
NAFLD (%)	17	20	27	40	< 0.0001
BMI (kg/m ²)	25.8±4.7	26.8±4.7	32.1±6.9	32.1±6./9	<0.0001
Physical activity index	9.1±1.9	9.0±1.9	8.7 ± 2.2	9.0 ± 2.0	0.28
Serum ALT (U/L)	17.9±21.1	18.1±13.9	18.7±13.9	19.7 ± 12.2	0.45
Triglycerides (mmol/l)	$1.27{\pm}1.20$	1.5 ± 1.14	1.27 ± 0.65	1.59±0.96	< 0.0001

	UDL shelesterel	1 25 0 22	1 22 0 21	1 21 0 20	1 21 0 24	-0.0001
	(mmol/l)	1.55±0.55	1.22±0.51	1.51±0.50	1.21±0.34	<0.0001
	Insulin (IU/L)	6.6±2.1	6.7±2.1	7.9±2.3	12.0±3.1	<0.0001
	(dose/day)	0.82 (0.03)	0.82 (0.06)	0.94 (0.18)	0.00 (0.10)	0.04
	Meat consumption (q/day)	141±85	146±84	149±80	131±69	0.42
	(g/uay)					
	Fruit consumption	160±146	174±84	139±110	190±149	0.25
	(g/day)					
_	Fish consumption	49±39	53±45	47±29	51±42	0.37
$\overline{}$	(g/day)					
	Vegetable consumption	165+129	169+133	142+98	234+208	0.06
J	(g/day)	100-127	10/_100	/0		
	Smoking (%)	14	14	25	18	0.05

* P-values from comparing across study groups

Model 1* Model 2[†] Model 3[‡] **Metabolic Group** OR (95 % CI) OR (95 % CI) OR (95 % CI) **P-value P-value** n/N % **P-value** (adjusted*) (adjusted*) (adjusted[†]) (adjusted[†]) (adjusted[‡]) (adjusted[‡]) Metabolic Group I 246/1,455 Reference Reference Reference 17 Metabolic Group II 82/402 1.23 (0.92-1.64) 0.16 1.35 (0.98-1.86) 1.06 (0.66-2.17) 20 0.062 0.75 Metabolic Group III 21/7927 1.95 (1.13-3.34) 0.016 2.22 (1.26-3.96) 0.0031 0.60 (0.31-1.16) 0.13 Metabolic Group IV 34/84 40 3.13 (1.94-5.06) < 0.0001 3.62 (2.15-6.09) < 0.0001 1.19 (0.66-2.16) 0.56

Table 2 Odds ratios (OR) and 95% confidence intervals (CI) of adult NAFLD according to youth metabolic phenotype.

Model 1* Adjusted for age and sex

Model 2[†] Further adjusted for adulthood alcohol consumption (portions per week), smoking (yes/no) and physical activity

Model 3‡ Further adjusted for adulthood BMI

Metabolic Group I = Reference (normal BMI, no metabolic disturbances), Group II (normal BMI, ≥ 1 metabolic disturbances), Group III

(overweight/obese, no metabolic disturbances), Group IV (overweight/obese, ≥ 1 metabolic disturbances).

Metabolic disturbances ($\geq 90^{\text{th}}$ % systolic blood pressure and diastolic pressure, $\geq 90^{\text{th}}$ % insulin $\geq 90^{\text{th}}$ %, triglycerides, $\leq 10^{\text{th}}$ percentile HDL-cholesterol)

n/N number of cases/total number of participants within the group

OR (95 % CI), Odds ratios 95% confidence intervals

Table 3 Odds ratios (OR) and 95% confidence intervals (CI) for adult NAFLD according to child metabolic syndrome (MetS) status

		Model 1*		Model 2†		Model 3‡	
Metabolic syndrome (1980)	n/N	OR (95 % CI adjusted*)	P-value (adjusted*)	OR (95 % CI adjusted†)	P-value (adjusted†)	OR (95 % CI adjusted‡)	P-value (adjusted‡)
No	271/1,534	Reference					
Yes	112/486	1.48 (1.14-1.92)	0.0031	1.57 (1.17-2.08)	0.0019	0.89 (0.64-1.24)	0.50

Model 1* Adjusted for age and sex

Model 2[†] Further adjusted for adulthood alcohol consumption (portions per week), smoking (yes/no) and physical activity

Model 3‡ Further adjusted for adulthood BMI

n=NAFLD cases in adulthood

N=Number of participants according to MetS status

Table 4 Odds ratios (OR) and 95% confidence intervals (CI) of adult NAFLD according to adiposity group in youth and adulthood comparing
A) normal BMI vs. overweight/obese i.e. a BMI of 25 kg/m² used as cut-off point. B) non-obese vs. obese i.e. BMI of 30 kg/m² cut-off point,

A)

Adiposity Group	n/N	Model 1*		Model 2**	
		OR (95 % CI)	P-value	OR (95 % CI)	P-value
Adiposity Group 1	48/858	Reference		Reference	
Adiposity Group 2	1/20	1.02 (0.13-7.97)	0.98	1.13 (0.14-8.92)	0.91
Adiposity Group 3	280/999	5.79 (4.17-8.03)	< 0.0001	6.14 (4.23-8.90)	< 0.0001
Adiposity Group 4	54/143	9.35 (5.91-14.80)	< 0.0001	11.3 (6.77-18.72)	< 0.0001

		Model 1*		Model 2**	
Adiposity Group	n/N	OR (95% CI)	P-value	OR (95% CI)	P-value
Adiposity Group 1	180/1,540	Reference		Reference	
Adiposity Group 2	6/68	0.82 (0.34-1.98)	0.67	0.89 (0.34-2.31)	0.80
Adiposity Group 3	148/317	7.86 (5.86-10.54)	< 0.0001	8.72 (5.86-12.10)	< 0.0001
Adiposity Group 4	49/95	8.07 (5.09-12.80)	< 0.0001	9.69 (5.86-16.01)	< 0.0001

n/N number of NAFLD cases/total number of participants in the group

Model 1*Age- and sex adjusted P-values for the comparison to Adiposity group 1 (reference)

Model 2**Further adjusted for adulthood alcohol consumption, physical exercise and smoking

Adiposity groups were: Adiposity group 1, normal childhood and adulthood BMI; group 2, high childhood, normal adulthood BMI; group 3, normal childhood BMI, high adulthood BMI, group 4, high childhood and adulthood BMI. Normal and high BMI were defined using international age-specific and sex-specific BMI cutoff points for overweight and obesity in childhood, and BMI cutoff points of 25 and 30 were used for adults

This article is protected by copyright. All rights reserved.

edte